Author's response to reviews

Title: Developing an online learning community for mental health professionals and service users: a discursive analysis

Authors:

Janet Smithson (j.smithson@exeter.ac.uk)
Ray B Jones (R.Jones-5@plymouth.ac.uk)
Emily Ashurst (Emily.Ashurst@plymouth.ac.uk)

Version: 4 Date: 21 February 2012

Author's response to reviews:

I have gone back over the article carefully and changed a couple of minor typos, but can't see any spelling errors.

Does the editor mean the quotations taken from the online forum? If so these are reproduced exactly as written (including typos and misspellings) by the participants. People typically make a lot of typos and also may deliberately use textspeak or abbreviations in online "talk", and this is relevant to our analysis, I don't think we can/should change the participants' words or spelling.